Author's response to reviews

Title: Protein induced by vitamin K absence or antagonist-II production is a strong predictive marker for extrahepatic metastases in early hepatocellular carcinoma: a prospective evaluation

Authors:

Hyun-Mi Bae (doctorbae75@naver.com)
Jeong-Hoon Lee (pindra@empas.com)
Jung-Hwan Yoon (yoonjh@snu.ac.kr)
Yoon Jun Kim (yoonjun@snu.ac.kr)
Dae Seog Heo (heo1013@snu.ac.kr)
Hyo-Suk Lee (hsleemd@snu.ac.kr)

Version: 7 Date: 8 October 2011

Author's response to reviews:

October 08, 2011

BMC cancer Editorial Office

Re: MS: 2096622635375850

Dear Editor.

We thank the editors and reviewers of the BMC cancer for giving us positive answers about our revision article entitled: “Protein induced by vitamin K absence or antagonist-II production is a strong predictive marker for extrahepatic metastases in early hepatocellular carcinoma: a prospective evaluation”.

We have made one correction in table 2 in the manuscript.

# We removed the horizontal line between Serum PIVKA-II† and Serum PIVKA-II (#300 mAU/mL) in table 2.

We hope the revised manuscript will better meet the requirements of the BMC cancer for publication. We thank again for the constructive comments and by the editors and reviewers.

Sincerely yours,

Thank you very much for your attention.

Sincerely yours,

Jung-Hwan Yoon, MD, PhD
Professor of Internal Medicine
Department of Internal Medicine,